Author’s response to reviews

Title: Association between Coronary Dominance and Acute Inferior Myocardial Infarction: a Matched, Case-control Study

Authors:
Li Wang (692748659@qq.com)
Jiamei Li (18792478857@163.com)
Ya Gao (gaoya090710@163.com)
Ruohan Li (674906954@qq.com)
Jingjing Zhang (zjflysun@163.com)
Dan Su (sudanbo@163.com)
Tao Wang (tonywong121@163.com)
Guang Yang (zjwcoconut@126.com)
Xiaochuang Wang (dr_xc_wang@163.com)

Version: 3 Date: 14 Jan 2019

Author’s response to reviews:

Responses for the Editor:

1. Guang Yang, Tao Wang and Dan Su’s initials are missing in the Author’s Contributions section. Please add them and ensure they qualify for authorship:

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. According to the ICMJE guidelines, to qualify as an author one should have:

a) made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; AND
b) been involved in drafting the manuscript or revising it critically for important intellectual content; AND

c) given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; AND

d) agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Response: Thank you for your suggestions. It was a mistake that I have forgotten to demonstrate their contributions in the manuscript. It was revised as shown in the section 'Authors’ contributions' in the manuscript.

2. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: Thank you for your suggestions.

The study was supported by the National Natural Science Foundation of China (81670049 and 81770057) and the National Key R&D Program in Shaanxi Province (2017SF-059). The funding Organizations were not involved in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

We have revised the section 'Funding' in the manuscript as above.
3. Please rename ‘Introduction’ to ‘Background’.

Response: We have renamed ‘Introduction’ to ‘Background’ in the manuscript.

4. Tracked changes removal: At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We will upload our manuscript as a single, final, clean version. All relevant tables/figures will also be clean versions. Figures will be uploaded as separate files.